---
displayed_sidebar: jsClientSidebar
---

import ParameterTypes from "@site/src/components/ParameterTypes"

# JS Client Reference

## Modules

- [addresses](modules/addresses.mdx)
- [admin](modules/admin.mdx)
- [admin/auth](modules/admin_auth.mdx)
- [admin/batch-jobs](modules/admin_batch_jobs.mdx)
- [admin/collections](modules/admin_collections.mdx)
- [admin/currencies](modules/admin_currencies.mdx)
- [admin/custom](modules/admin_custom.mdx)
- [admin/customer-groups](modules/admin_customer_groups.mdx)
- [admin/customers](modules/admin_customers.mdx)
- [admin/discounts](modules/admin_discounts.mdx)
- [admin/draft-orders](modules/admin_draft_orders.mdx)
- [admin/gift-cards](modules/admin_gift_cards.mdx)
- [admin/inventory-item](modules/admin_inventory_item.mdx)
- [admin/invites](modules/admin_invites.mdx)
- [admin/notes](modules/admin_notes.mdx)
- [admin/notifications](modules/admin_notifications.mdx)
- [admin/order-edits](modules/admin_order_edits.mdx)
- [admin/orders](modules/admin_orders.mdx)
- [admin/payment-collections](modules/admin_payment_collections.mdx)
- [admin/payments](modules/admin_payments.mdx)
- [admin/price-lists](modules/admin_price_lists.mdx)
- [admin/product-categories](modules/admin_product_categories.mdx)
- [admin/product-tags](modules/admin_product_tags.mdx)
- [admin/product-types](modules/admin_product_types.mdx)
- [admin/products](modules/admin_products.mdx)
- [admin/publishable-api-keys](modules/admin_publishable_api_keys.mdx)
- [admin/regions](modules/admin_regions.mdx)
- [admin/reservations](modules/admin_reservations.mdx)
- [admin/return-reasons](modules/admin_return_reasons.mdx)
- [admin/returns](modules/admin_returns.mdx)
- [admin/sales-channels](modules/admin_sales_channels.mdx)
- [admin/shipping-options](modules/admin_shipping_options.mdx)
- [admin/shipping-profiles](modules/admin_shipping_profiles.mdx)
- [admin/stock-locations](modules/admin_stock_locations.mdx)
- [admin/store](modules/admin_store.mdx)
- [admin/swaps](modules/admin_swaps.mdx)
- [admin/tax-rates](modules/admin_tax_rates.mdx)
- [admin/uploads](modules/admin_uploads.mdx)
- [admin/users](modules/admin_users.mdx)
- [admin/variants](modules/admin_variants.mdx)
- [auth](modules/auth.mdx)
- [carts](modules/carts.mdx)
- [collections](modules/collections.mdx)
- [customers](modules/customers.mdx)
- [gift-cards](modules/gift_cards.mdx)
- [index](modules/index.mdx)
- [line-items](modules/line_items.mdx)
- [order-edits](modules/order_edits.mdx)
- [orders](modules/orders.mdx)
- [payment-collections](modules/payment_collections.mdx)
- [payment-methods](modules/payment_methods.mdx)
- [product-categories](modules/product_categories.mdx)
- [product-tags](modules/product_tags.mdx)
- [product-types](modules/product_types.mdx)
- [product-variants](modules/product_variants.mdx)
- [products](modules/products.mdx)
- [regions](modules/regions.mdx)
- [return-reasons](modules/return_reasons.mdx)
- [returns](modules/returns.mdx)
- [shipping-options](modules/shipping_options.mdx)
- [swaps](modules/swaps.mdx)
